Author’s response to reviews

Title: “Whatever Happens, Happens” Challenges of End-of-Life Communication from the Perspective of Older Adults and Family Caregivers: A Qualitative Study

Authors:

Jennifer Im (jennifer.im@mail.utoronto.ca)
Susanna Mak (susanna.mak@sinahealthsystem.ca)
Ross Upshur (ross.upshur@gmail.com)
Leah Steinberg (leah.steinberg@sinahealthsystem.ca)
Kerry Kuluski (Kerry.Kuluski@thp.ca)

Version: 2 Date: 05 Nov 2019

Author’s response to reviews:

Dear reviewers,

There were four comments in this round of revisions. Specifically, the following 4 items were addressed:

1) The ethics statement in the manuscript explicitly states that written informed consent were obtained from participants to participate in the study which includes participating in a recorded interview, and permission to publish de-identified material in publications and presentations.

2) The ethics statements explicitly states that we obtained written consent to publish de-identified findings in publications and presentations.

3) In the manuscript, "findings" was changed to "results."

4) All uploaded documents are "clean" versions that do not contain tracked changes. Tables and supplementary files have been uploaded separately.

Thank you,

Jennifer Im